food & nutrition (

REVIEW ARTICLE

Effects of ketogenic diet on weight loss parameters among obese or overweight patients with polycystic ovary syndrome: a systematic review and meta-analysis of randomized controlled trails

Nan-nan Xing* 💿, Fang Ren 💿 and Hui Yang 💿

Department of Obstetrics and Gynecology, Shengjing Hospital of China Medical University, Shenyang, China

Popular scientific summary

- This meta-analysis of 11 RCTs examined how a low-carbohydrate ketogenic diet (KD) affects weight loss factors, such as weight, WC, BMI, and fat mass (FM), in women with PCOS who were obese or overweight.
- The pooled results indicated that the KD intervention led to significant reductions in body weight, BMI, WC, and FM.
- Further RCTs with larger sample sizes are necessary to authenticate the evidence and explore the most favorable dietary regimens.

Abstract

Aim: To evaluate how effective a low carbohydrate ketogenic diet (KD) is for changing key physical measurements such as weight, waist circumference (WC), body mass index (BMI), and fat mass (FM) in women with polycystic ovary syndrome (PCOS) who were obese or overweight.

Methods: Several online databases, including PubMed, Scopus, EMBASE, Cochrane Library, and Web of Science (WOS), were searched systematically to find relevant randomized controlled trials (RCTs) up until June 2023. The Q-test and I^2 statistics were used to assess the level of heterogeneity among the included studies. The data were then combined using either a fixed or random effects model and presented as a weighted mean difference (WMD) along with a 95% confidence interval (CI).

Results: Of the 682 citations, 11 RCTs were included. The pooled results showed a significant decrease in the WMD of weight levels [WMD = -9.13 kg; 95% CI, -11.88, -6.39, P < 0.001; $I^2 = 87.23\%$] following KD. Moreover, KD significantly reduced BMI levels [WMD = -2.93 kg/m^2 ; 95% CI, -3.65, -2.21, P < 0.001; $I^2 = 78.81\%$] compared to the controls. Patients with PCOS received KD demonstrated significant decrease in WC [WMD = -7.62 cm; 95% CI, -10.73, -4.50, P < 0.001; $I^2 = 89.17\%$] and FM [WMD = -5.32 kg; 95% CI, -7.29, -3.36, P < 0.001; $I^2 = 83.97\%$].

Conclusion: KD was associated with lower weight loss (WL) parameters, including weight, BMI, WC, and FM, in obese or overweight women with PCOS, highlighting the significance of physicians and nurses in taking care of the nutritional needs of overweight/obese patients with PCOS.

Keywords: ketogenic diet; weight loss; obesity; polycystic ovary syndrome; meta-analysis.

To access the supplementary material, please visit the article landing page

Received: 25 July 2023; Revised: 13 September 2023; Accepted: 21 September 2023; Published: 12 March 2024

Polycystic ovary syndrome (PCOS) is a common, complex hormonal and metabolic condition of unknown cause. It affects 5–20% of women of reproductive age worldwide depending on how it is diagnosed (1, 2). Women with PCOS typically exhibit oligoanovulation, biochemical or clinical hyperandrogenism, and/or polycystic ovarian morphology. Consequently, patients may present with symptoms such as infertility, insulin resistance (IR), hirsutism, acne, and dyslipidemia (3). Patients with PCOS are at risk for a range of medical complications throughout their lives, including obesity, metabolic dysfunction, vascular dysfunction, malignancy, reproductive complications, and mood disorders (4).

The treatment of PCOS typically involves a series of lifestyle modifications such as adjustments in diet, physical exercise, and weight management. Of these measures, weight loss (WL) is considered one of the most efficient strategies for regulating the menstrual cycle and alleviating PCOS symptoms (5, 6). The ketogenic diet (KD) is a dietary approach distinguished by its low-carbohydrate, high-fat, and moderate protein composition, which induces a state of nutritional ketosis through the elevated production of ketones, predominantly β -hydroxybutyrate and acetoacetate. Notably, KD has been found to have anti-seizure properties and is used to treat epilepsy that is difficult to manage (7). Research suggests that it may also be a useful treatment for other neurological conditions such as Parkinson's disease, autism, and Alzheimer's disease (8).

Moreover, previous clinical trials have reported beneficial therapeutic effects of KD in women diagnosed with PCOS by decreasing androgen secretion, elevating the circulating levels of sex hormone-binding globulin (SHBG), and improving insulin sensitivity (9). Furthermore, the endocrine and metabolic effects of KD are primarily characterized by a considerable decline in fasting serum insulin, percentage of free testosterone, weight, and luteinizing hormone (LH)/follicle-stimulating hormone (FSH) ratio in females with PCOS and obesity (10). An exceptional effect of KD in PCOS is the activation of AMPK and SIRT1, despite not being a caloric deprivation diet (11, 12). Upon activation, both AMPK and SIRT1 positively influence glucose homeostasis and enhance insulin sensitivity (12, 13).

Several meta-analyses have evaluated the impact of nutritional interventions on various aspects of PCOS, including biochemical factors (14–16), androgenic profiles (17, 18), and IR (19–22). However, no published meta-analysis has assessed the effect of KD on WL parameters in obese or overweight patients with PCOS. Therefore, this meta-analysis aimed to evaluate the efficacy of KD on WL parameters, including weight, waist circumference (WC), body mass index (BMI), and fat mass (FM), in obese or overweight women with PCOS.

Method

Literature search strategy

The PRISMA guidelines were followed in this systematic review and meta-analysis (23). We searched online databases such as PubMed, Web of Science (WOS), Scopus, Cochrane, and EMBASE to find relevant literature on the impact of the KD (carbohydrates less than 50 g/day) on WL parameters. We focused on clinical trials published until June 2023. For the initial data gathering, we used the following search strategy: ('Diet, Ketogenic' [MeSH] OR 'Ketogenic*'[Title/Abstract]) AND ('Polycystic Ovary Syndrome' [MeSH] OR 'Polycystic Ovary Syndrome' [Title/ Abstract] OR PCOS'[Title/Abstract]) AND ('Weight Loss' [MeSH] OR 'Weight Loss*'[Title/Abstract] OR 'Body Mass Index'[MeSH] OR 'Body Mass Index'[Title/ Abstract] OR 'BMI'[Title/Abstract] OR 'Obesity'[MeSH] OR 'Obesity'[Title/Abstract] OR 'Overweight'[MeSH] OR 'Overweight'[Title/Abstract] OR 'waist circumference'[MeSH] OR 'waist circumference'[Title/Abstract] OR 'Body Mass*'[Title/Abstract] OR 'Fat Mass'[Title/ Abstract]). A manual search of the references in the selected articles and previous reviews was conducted to identify additional relevant articles. The titles and abstracts of the included trials, data extraction, and evaluation of eligible studies were conducted separately by two authors (FR and HY). Any discrepancies were resolved by consultation with a third author (NX) to ensure accuracy.

Criteria for inclusion and exclusion

This review included clinical trials that met the following criteria: 1) trials with crossover, parallel, or before-after designs involving overweight or obese women with PCOS; 2) trials that investigated the effects of KDs (carbohydrates less than 50 g/day) on WL parameters, such as WL, BMI, WC, and FM; and 3) articles published in English. Trials that did not report mean differences in WL parameters along with standard deviation (SD), *in vitro* studies, abstracts of seminars without full text, or case reports or letters to editors were excluded from the current meta-analysis.

Extraction of data and quality assessment

Two reviewers individually conduct the quality assessments and data extraction for the included trials. The National Institutes of Health (NIH) and the Cochrane Collaboration Risk of Bias tools were used for quality assessment of the included trials with RCT and before-after design, while a standard Excel sheet was used for data extraction. The methodological quality of the clinical trials included in this review was assessed based on criteria such as 'blinding of participants and outcome assessors, randomization generation, allocation concealment, selective outcome reporting, incomplete outcome data, and other sources of bias'. The following data were extracted: first author's name, age of participants, study location, publication year, study method, total sample size, type and duration of intervention, and mean (SD) changes in WL factors (WL, BMI, WC, and FM) for women with PCOS with and without KDs.

Statistical analysis

STATA V.11.0 and Comprehensive Meta-Analysis V2 (CMA V2) were used for all statistical analyses. The effects of KDs on changes in WL factors were

determined, and pooled effect sizes were calculated using weighted mean differences (WMDs) with a 95% confidence interval (CI). The following formulae were used for the calculations: [meanpost - meanpre] for the mean change and $\left[\sqrt{(SDpost2 + SDpre2)} - [2 \times R \times R\right]$ $SDpost \times SDpre$])] for the corresponding SD. A correlation coefficient with a value of 0.8 was postulated for the variable 'R' in the computation. The evaluation of heterogeneity within the studies incorporated was conducted through the application of statistical tests, namely, the Cochran (Q) and I-squared (I^2) tests. If the value of I^2 surpassed 50% and a P-value less than 0.1 was obtained, then the application of the random-effects model was applied. Alternatively, if the aforementioned criteria were not met, then the fixed-effects model was employed. Subgroup analyses were done to detect possible sources of heterogeneity based on the moderator variables. Sensitivity analyses were performed to evaluate the impact of each trial on the dependability of the combined WMDs by means of the leave-oneout method. Quantitative and visual assessments were

conducted to evaluate potential evidence of publication bias using funnel plot and Egger's test. The level of statistical significance was established as P < 0.05.

Results

After eliminating duplicate and irrelevant studies, 11 articles (comprising 12 trials) were selected for the present meta-analysis out of the initial 349 citations. The systematic procedure of identifying relevant studies is illustrated in Fig. 1.

Among the included articles, three were randomized, double-blind, placebo-controlled trials, whereas the remaining eight were before-after studies. This meta-analysis included 426 women with PCOS. Yang et al. (24) conducted a study that examined different disease types, specifically PCOS with non-hyperuricemia and hyperuricemia, and treated each disease type as a separate study within the meta-analysis. The intervention duration across the included studies varied from 6 to 24 weeks. The selected trials were published between 2005 and 2023. Table 1 elucidates the characteristics of the included trials.



Fig. 1. Flowchart study identify and study selection.

Table 1. Main characteristic of included studies

Authors/ country	Study design	Total sample size	Interventions	Duration (weeks)	Outcomes
Li et al. (25)/China	RCT	18	The KD was composed of approximately 5–10% energy derived from carbohydrates, which equates to a maximum of 50 g/day. Protein intake accounted for 18–27% of energy, while the majority of energy, approximately 70–75%, was obtained from fat.	12	WL, BMI (Patients: 29.82 ± 2.36; Control: 32.81 ± 3.75), and FM
Avolio et al. (26)/Italy	Before-after trial	57	A ketogenic diet was defined as a carbohydrate intake below 20 g/day. The ketogenic diet aimed to achieve a daily energy intake of 750–800 calories, with $35-40\%$ of the calories coming from proteins (equivalent to 1.2 g per kilogram of ideal body weight). Fat intake accounted for $45-50\%$ of calories, with less than 10% of calories derived from saturated fat. Carbohydrate intake was limited to 10% of the total calories, which equated to less than 20 g/day.	24	FM
Cincione et al. (27)/Italy	Before-after trial	17	The maximum daily carbohydrate intake was limited to 30 g/day. Additionally, the total daily caloric intake was set at approximately 600 calories. The diet employed in the study was classified as a very low-calorie ketogenic diet (VLCKD).	6	WL, BMI (31.84 ± 5.85 kg/m ²), WC, and FM
Cincione et al. (28)/Italy	RCT	144	The daily intake of carbohydrates was restricted to a maxi- mum of 30 g/day, and the total daily caloric intake was set at approximately 600 calories. The diet implemented in this study was categorized as a very low-calorie ketogenic diet (VLCKD).	6	WL, BMI (KD: 34.69 ± 5.26; Mediterranean: 33.96 ± 4.72),WC, and FM
Mavropoulos et al. (29)/USA	Before-after trial	5	Throughout the 6-month study period, a low-carbohydrate ketogenic diet (LCKD) was implemented, with an intake of less than 20 g of carbohydrates per day, as tolerated by the participants.	24	WL
liran et al. (30)/China	Before-after trial	10	The ketogenic diet (KD) followed a daily carbohydrate intake of less than 50 g.	4	WL and BMI (27.1 ± 3.0)
Jian et al. (31)/China	Before-after trial	51	The KD included a daily carbohydrate intake of 50 g or less.	4	WL and BMI (29.87 ± 3.79)
Pandurevic et al. (32)/Italy	RCT	30	During the very-low-calorie ketogenic diet (VLCKD) phase, there were three steps, with daily caloric intake ranging from 600 to 800 kcal. The diet primarily included high-biologi- cal-value protein preparations derived from sources such as cow milk, soy, eggs, green peas, and cereals. Carbohydrate intake obtained from vegetables was restricted to less than 50 g/day. Additionally, the diet included a daily intake of 10 g olive oil.	16	BMI (Experimental: 33.9 ± 3.8; Control 33.7 ± 4.3), WC, and FM
Magagnini et al. (33)/Italy	Before-after trial	25	In the VLCKD protocol, the diet includes high-biological pro- teins obtained from sources, such as milk, peas, and whey. Each protein preparation, with an approximate caloric content of 100-150 kcal, contained 18 g of protein, 4 g of carbohydrates, and 3 or 4 g of fat. This composition was consistent with the VLCKD requirements.	12	BMI (32.8 ± 1.0) and WC
Paoli et al. (12)/Italy	Before-after trial	14	A modified KD protocol was employed, which involved the use of formulas with high protein content (19 g per portion) and very low carbohydrate content (3.5 g per portion). This modified protocol aimed to maintain the desired macronutrient ratios for effective ketogenic diet implementation.	12	WL, BMI (28.84 ± 2.10), WC, and FM
Yang (a) et al. (24)/China	Before-after trial	27	The ketogenic diet had a recommended carbohydrate intake of less than 50 g/day. The energy supply ratio for the three macronutrients in the ketogenic diet was approximately 70–75% fat, 5–10% carbohydrate, and 18–27% protein.	12	WL
Yang (b) et al. (24)/China	Before-after trial	28	The ketogenic diet had a recommended carbohydrate intake of less than 50 g/day. The energy supply ratio for the three macronutrients in the ketogenic diet was approximately 70–75% fat, 5–10% carbohydrate, and 18–27% protein.	12	WL

Main outcomes

Figure 2 shows forest plots illustrating the effects of KDs on the WL parameters. The pooled meta-analysis, utilizing a random-effect model, revealed significant reductions in weight (WMD = -9.13 kg; 95% CI, -11.88 to -6.39; P < 0.01; I^2 : 87.23%), BMI (WMD = -2.93 kg/m^2 ; 95% CI, -3.65 to -2.21; P < 0.01; I^2 : 78.81%), WC (WMD = -7.62 cm; 95% CI, -10.73 to -4.50; P < 0.01; I^2 : 89.17%), and FM (WMD = -6.62 kg; 95% CI, -8.44 to -4.80; P < 0.01; I^2 : 53.92%).

Sensitivity analyses and subgroup analysis

A)

Following the sensitivity analysis, our findings exhibited that there were no significant deviations between the preand post-sensitivity WMDs for variables such as weight, WC, BMI, and FM. However, when excluding the study by Jian et al. (31), the sensitivity analysis showed a lower pooled WMD of -9.69 (95% CI: -12.45, -6.93) for weight, and when excluding the study by Yang (b) et al. (24), the higher pooled WMD was -8.12 (95% CI: -10.28, -5.96). For BMI, the lower pooled WMD in sensitivity analysis was -3.17 (95% CI: -3.79, -2.55) after excluding Jian et al. (31), and the higher was -2.68 (95% CI: -3.27, -2.09) after removing Magagnini et al. (33). Regarding WC and FM, the lower pooled WMD in sensitivity analysis was -9.23 (95% CI: -9.99, -8.46) and -7.78(95% CI: -9.10, -6.47), respectively, after excluding studies by Paoli et al. (12) and Avolio et al. (26). The higher pooled WMDs in sensitivity analysis were -6.98 (95% CI: -10.51, -3.46) and -6.42 (95% CI: -8.51, -4.34) after

Study				W with 9	MD 95% C		Weight (%)
Cincione et al. (2023)		-		-8.36 [-1	6.44,	-0.28]	6.55
Li et al. (2021)				-10.30 [-1	6.00,	-4.60]	9.13
Cincione et al. (2021)				-5.63 [-	9.67,	-1.59]	11.37
Mavropoulos et al. (2005)			-	-12.33 [-2	4.32,	-0.34]	3.89
Jian et al. (2022)			-	5.79 [-	7.69,	-3.89]	14.05
Paoli et al. (2020)				-9.43 [-1]	2.08,	-6.78]	13.22
liran et al. (2019)				5.90 [8.37,	-3.43]	13.43
Bai (a) et al. (2022)			-	-11.20 [-1	2.94,	-9.46]	14.20
Bai(b) et al. (2022)		-	-	-14.70 [-1	6.48,	-12.92]	14.17
Overall Hotorogonoity: $r^2 = 12.04 I^2 = 87.22\% H^2 = 7.92$			•	-9.13 [-1	1.88,	-6.39]	
Therefore $O(R) = 02.023, R = 7.00$,						
Test of $\theta_i = \theta_j$: Q(8) = 62.63, $\beta = 0.00$							
lest of $\theta = 0$: z = -6.52, p = 0.00	30	-20	-10				
Random-effects DerSimonian-Laird model	-30	-20	-10	U			





Fig. 2. The effects of a ketogenic diet on (A) weight, (B) BMI, (C) WC, and (D) FM levels in overweight or obese women with PCOS.

C)

Study					WMD with 95% CI		Weight (%)
Cincione et al. (2023)		-			7.22 [-15.88, 1.4	4]	8.79
Pandurevic et al. (2023)					-8.40 [-13.06, -3.7	4]	17.00
Cincione et al. (2021)		_			-9.44 [-12.62, -6.2	6]	21.38
Magagnini et al. (2022)		-			-9.26 [-10.06, -8.4	6]	27.00
Paoli et al. (2020)					-4.01 [-5.52, -2.5	0]	25.82
Overall					-7.62 [-10.73, -4.5	[0]	
Heterogeneity: $\tau^2 = 9.18$, $I^2 = 89.17\%$, $H^2 = 9.23$							
Test of $\theta_i = \theta_j$: Q(4) = 36.93, p = 0.00							
Test of θ = 0: z = -4.79, p = 0.00							
	-15	-10	-5	0			

Random-effects DerSimonian-Laird model

D)



Random-effects DerSimonian-Laird model

Fig. 2. Continued The effects of a ketogenic diet on (A) weight, (B) BMI, (C) WC, and (D) FM levels in overweight or obese women with PCOS.

removing studies by Magagnini et al. (33) and Cincione et al. (27), respectively.

Figure 3A–D depicts the outcomes of the sensitivity analysis conducted subsequent to the exclusion of each trial from the meta-analysis. When we conducted a subgroup analysis according to the duration of the intervention (\geq 12 weeks vs. <12 weeks), noteworthy reductions in WL factors were observed in studies with intervention durations of \geq 12 weeks and <12 weeks.

Publication bias and quality assessment

Upon performing quantitative and visual assessments, we did not find any significant evidence of publication bias. However, the P-values obtained from Egger's test were 0.23 for weight, 0.96 for BMI, 0.91 for WC, and 0.76 for FM (Supplementary Fig. S1–S4). All included trials with RCT design were of low quality using the Cochrane Collaboration Risk of Bias tool, and 3 of 5 before and after articles were of high quality using the NIH tool (Supplementary Tables S1–S2).

Discussion

To the best of the authors' knowledge, this is the first systematic review and meta-analysis that investigated the effect of KDs on body composition in obese or overweight PCOS patients over 11 included RCTs. Pooled findings revealed that body weight, BMI, WC, and FM significantly decreased following KD intervention.

Research is growing on how adjusting diets can help treat PCOS. When carbohydrate intake is reduced and fat supply is improved, the body will start metabolizing fat for energy. KDs mimic the metabolic state of fasting and have been proven in several RCTs to decrease obesity in participants. Body weight, BMI, FM, and WC could be significantly reduced (34–36). Patients diagnosed with non-alcoholic fatty liver disease (NAFLD) who followed a KD also lost significant amounts of weight, with reductions in visceral adipose tissue and liver fat fraction compared to those on a standard diet (37). A recent meta-analysis of 10 RCTs assessed the effect of KD on body composition and metabolic parameters in



A)

B)

Fig. 3. Sensitivity analyses findings of the effects of a ketogenic diet on (A) weight, (B) BMI, (C) WC, and (D) FM levels in overweight or obese women with PCOS.

-2.93

patients with cancer. Their pooled effect sizes demonstrated a significant decrease in body weight and FM post KDs intervention (38). KDs could potentially aid in reducing body fat and visceral adipose tissue without affecting lean body mass (39, 40). Moreover, a recently published meta-analysis of 21 clinical trials involving obese/overweight patients with T2DM found that the implementation of a low-carbohydrate KD resulted in significant decreases in body weight, BMI, and WC. Additionally, non-diabetic patients experienced significant decreases in body FM. The findings of this study suggest that such KDs could be beneficial for obese and overweight patients with T2DM, as it can improve cardiovascular risk factors such as weight, glycemic, and

liran et al. (2019)

-3.81 -3.65

lipid profile (41). Our meta-analysis has established that KDs have the potential to reduce the weight, WC, BMI, and FM of patients with PCOS.

-2.21

-1.92

PCOS often leads to weight gain and central obesity, which typically appear before the onset of anovulatory menstrual cycles (42). In these cases, visceral adiposity is linked to increased IR, which worsens both metabolic and reproductive disorders. Obesity also has a significant effect on the PCOS phenotype as it is linked to a greater occurrence of menstrual irregularities, hyperandrogenemia, and hirsutism (26, 43). The etiology of PCOS remains uncertain; however, IR appears to be a significant contributing factor. Women diagnosed with PCOS frequently exhibit concomitant IR, which can be attributed

Meta-analysis estimates, given named study is omitted Lower CL imit GEstimate Upper CI Limit Cincione et al. (2023) Pandurevic et al. (2023) Cincione et al. (2021) Magagnini et al. (2022) Paoli et al. (2020) -7.62 -4 50 -3 34 -11:10473 Meta-analysis estimates, given named study is omitted Lower CLL imit ○ Estimate Upper CI Limit Cincione et al. (2023)



C)



Fig. 3. Continued Sensitivity analyses findings of the effects of a ketogenic diet on (A) weight, (B) BMI, (C) WC, and (D) FM levels in overweight or obese women with PCOS.

to excessive weight and obesity (26). Recent reproductive and endocrine studies have highlighted a significant role of diet in IR. Studies have suggested that specific well-balanced diets, including low glycemic index (GI) diets, Mediterranean diets, Dietary Approaches to Stop Hypertension (DASH) diets, and vegetarian diets, can improve IR, regulate metabolism, manage body weight, and prevent related complications (44–46). Despite these recommendations, PCOS patients are often hesitant to follow the guidelines, indicating the importance of dietary and exercise interventions as first-line management for these patients (47), or adopting self-help methods (48, 49).

Adipose tissue serves as an additional source of androgens, and its excessive accumulation can exacerbate

hyperandrogenism (50). Adipocytes located in the abdominal region exhibit greater activity as endocrine cells than those found in the lower body, which characterizes gynoid obesity (51). These cells have a greater sensitivity to catecholamines and a reduced sensitivity to insulin, leading to an increase in insulin production and mild inflammation, changes in lipid levels, higher levels of androgen production, and lower levels of SHBG. These factors collectively promote anovulation (12). Women with PCOS who exhibit hyperandrogenism, high BMI, and IR have diminished metabolic flexibility (as determined by changes in respiratory quotient following insulin stimulation) (52, 53). The present meta-analysis also demonstrated a significant reduction in WL and BMI in PCOS patients.

Guidelines for treating PCOS recommend lifestyle changes as an initial therapeutic approach. This involves modifying diet and increasing physical activity, which have been found to be effective in achieving WL and altering body composition (27). The present meta-analysis confirms previous reports that a low-calorie diet with a high nutritional value is effective for managing PCOS. Most of the included studies reported a carbohydrate intake lower than 20 g in the KD, which improves insulin sensitivity by reducing carbohydrates and calories. KD therapy also personalized protein intake to maintain lean mass for sustained results (54, 55). The lipid supply is rich in mono- and polyunsaturated fatty acids that are known for their insulin-sensitizing effects, as well as their anti-inflammatory actions (56). However, the long-term effects of this therapy need to be confirmed through follow-up studies to determine whether sustained results can be achieved with less restrictive, more sustainable eating behaviors.

Our meta-analysis differs from previous studies because it included more newly published studies and statistics. This meta-analysis had certain limitations. First, blinding of the outcome assessment was not possible because dietary modification was the only intervention. Moreover, dietary regulation was not strictly implemented, and ingredients varied, which may have adversely affected the accuracy of the results. Second, heterogeneity existed in our analysis owing to differences in patient physical characteristics. Third, a potential limitation of this study is that some of the studies evaluated had a before-andafter design without a control group, which may affect the reliability of the overall results. Finally, although we gathered a substantial number of RCTs, these are still insufficient to reach a definitive conclusion on the benefits of KDs. It is essential to conduct more RCTs to support our findings, especially since most studies have focused on interventions lasting less than 6 months. Future research should examine whether these effects can be maintained over a longer period. Moreover, given the proficiency and responsibility of nurses in catering to the nutritional demands of patients, it becomes imperative for them to engage in nutrition screening and offer suitable dietary guidance, such as the KD, to enhance the adoption of wholesome eating habits and consequently ameliorate the body weight outcomes of obese or overweight patients with PCOS.

Conclusion

Dietary modifications may contribute to the amelioration of weight management in females afflicted with PCOS. KD might be considered an ideal option for enhancing weight management, as it was shown by decreasing body weight, BMI, WC, and FM in obese or overweight PCOS patients. Overall, dietary modifications constitute an efficacious, tolerable, and secure methodology, thereby providing viable alternatives for patients who are unable to cope with the gastrointestinal side effects of metformin. Nevertheless, given the limited number of studies and the small sample sizes observed in certain included studies, it is imperative to interpret the findings cautiously. Additional RCTs featuring robust designs and large sample sizes are warranted to validate the evidence and investigate optimal dietary patterns.

Acknowledgments

None.

Conflict of interests and funding

No potential conflict of interest was reported by the authors.

Authors' contributions

Nan-nan Xing designed the work. Nan-nan Xing, Fang Ren, and Hui Yang conducted the work and collected and analyzed the data. Nan-nan Xing, Fang Ren, and Hui Yang drafted the manuscript and revised it critically. The manuscript has been thoroughly read and approved by all authors.

Data availability

All data generated or analyzed during this study are included in this article.

Ethics approval

None.

References

- Azziz R, Carmina E, Chen Z, Dunaif A, Laven JS, Legro RS, et al. Polycystic ovary syndrome. Nat Rev Dis Prim 2016; 2(1): 1–18. doi: 10.1038/nrdp.2016.57
- Azziz R. Polycystic ovary syndrome. Obstetr Gynecol 2018; 132(2): 321–36. doi: 10.1097/AOG.00000000002698
- Yau TT, Ng NY, Cheung L, Ma RC. Polycystic ovary syndrome: a common reproductive syndrome with long-term metabolic consequences. Hong Kong Med J 2017; 23(6): 622. doi: 10.12809/hkmj176308
- Kakoly N, Khomami M, Joham A, Cooray S, Misso M, Norman R, et al. Ethnicity, obesity and the prevalence of impaired glucose tolerance and type 2 diabetes in PCOS: a systematic review and meta-regression. Hum Reprod Update 2018; 24(4): 455–67. doi: 10.1093/humupd/dmy007
- Lim SS, Hutchison SK, Van Ryswyk E, Norman RJ, Teede HJ, Moran LJ. Lifestyle changes in women with polycystic ovary syndrome. Cochrane Database Syst Rev 2019; 3: CD007506. doi: 10.1002/14651858.CD007506.pub4
- Saadati N, Haidari F, Barati M, Nikbakht R, Mirmomeni G, Rahim F. The effect of low glycemic index diet on the reproductive and clinical profile in women with polycystic ovarian syndrome: a systematic review and meta-analysis. Heliyon 2021; 7(11): e08338. doi: 10.1016/j.heliyon.2021.e08338

- Joshi S, Ostfeld RJ, McMacken M. The ketogenic diet for obesity and diabetes – enthusiasm outpaces evidence. JAMA Intern Med 2019; 179(9): 1163–4. doi: 10.1001/jamainternmed.2019.2633
- Che X, Chen Z, Liu M, Mo Z. Dietary interventions: a promising treatment for polycystic ovary syndrome. Ann Nutr Metabol 2021; 77(6): 313–23. doi: 10.1159/000519302
- Barrea L, Verde L, Camajani E, Cernea S, Frias-Toral E, Lamabadusuriya D, et al. Ketogenic diet as medical prescription in women with polycystic ovary syndrome (PCOS). Curr Nutr Rep 2023; 12(1): 56–64. doi: 10.1007/s13668-023-00456-1
- Gupta L, Khandelwal D, Kalra S, Gupta P, Dutta D, Aggarwal S. Ketogenic diet in endocrine disorders: current perspectives. J Postgrad Med 2017; 63(4): 242. doi: 10.4103/jpgm. JPGM_16_17
- Draznin B, Wang C, Adochio R, Leitner J, Cornier M-A. Effect of dietary macronutrient composition on AMPK and SIRT1 expression and activity in human skeletal muscle. Hormone Metabol Res 2012; 44(09): 650–5. doi: 10.1055/s-0032-1312656
- Paoli A, Mancin L, Giacona MC, Bianco A, Caprio M. Effects of a ketogenic diet in overweight women with polycystic ovary syndrome. J Transl Med 2020; 18(1): 1–11. doi: 10.1186/ s12967-020-02277-0
- Ruderman N, Julia Xu X, Nelson L, Cacicedo J, Saha A, Lan F, et al. AMPK 615 and SIRT1: a long-standing partnership. Am J Physiol Endocrinol Metab 2010; 298(4): E751–60. doi: 10.1152/ ajpendo.00745.2009
- Abdolahian S, Tehrani FR, Amiri M, Ghodsi D, Yarandi RB, Jafari M, et al. Effect of lifestyle modifications on anthropometric, clinical, and biochemical parameters in adolescent girls with polycystic ovary syndrome: a systematic review and meta-analysis. BMC Endocrine Disord 2020; 20(1): 1–17. doi: 10.1186/ s12902-020-00552-1
- Akbari M, Ostadmohammadi V, Lankarani KB, Tabrizi R, Kolahdooz F, Heydari ST, et al. The effects of vitamin D supplementation on biomarkers of inflammation and oxidative stress among women with polycystic ovary syndrome: a systematic review and meta-analysis of randomized controlled trials. Hormone Metabol Res 2018; 50(04): 271–9. doi: 10.1055/s-0044-101355
- Heydarpour F, Hemati N, Hadi A, Moradi S, Mohammadi E, Farzaei MH. Effects of cinnamon on controlling metabolic parameters of polycystic ovary syndrome: a systematic review and meta-analysis. J Ethnopharmacol 2020; 254: 112741. doi: 10.1016/j.jep.2020.112741
- Azadi-Yazdi M, Nadjarzadeh A, Khosravi-Boroujeni H, Salehi-Abargouei A. The effect of vitamin D supplementation on the androgenic profile in patients with polycystic ovary syndrome: a systematic review and meta-analysis of clinical trials. Hormone Metabol Res 2017; 49(03): 174–9. doi: 10.1055/s-0043-103573
- Carmina E, Azziz R, Bergfeld W, Escobar-Morreale HF, Futterweit W, Huddleston H, et al. Female pattern hair loss and androgen excess: a report from the multidisciplinary androgen excess and PCOS committee. J Clin Endocrinol Metabol 2019; 104(7): 2875–91. doi: 10.1210/jc.2018-02548
- Heshmati J, Omani-Samani R, Vesali S, Maroufizadeh S, Rezaeinejad M, Razavi M, et al. The effects of supplementation with chromium on insulin resistance indices in women with polycystic ovarian syndrome: a systematic review and meta-analysis of randomized clinical trials. Hormone Metabol Res 2018; 50(03): 193–200. doi: 10.1055/s-0044-101835
- 20. Łagowska K, Bajerska J, Jamka M. The role of vitamin D oral supplementation in insulin resistance in women with polycystic ovary syndrome: a systematic review and meta-analysis of

randomized controlled trials. Nutrients 2018; 10(11): 1637. doi: 10.3390/nu10111637

- Shang Y, Zhou H, Hu M, Feng H. Effect of diet on insulin resistance in polycystic ovary syndrome. J Clin Endocrinol Metabol 2020; 105(10): 3346–60. doi: 10.1210/clinem/dgaa425
- 22. Sadeghi A, Djafarian K, Mohammadi H, Shab-Bidar S. Effect of omega-3 fatty acids supplementation on insulin resistance in women with polycystic ovary syndrome: meta-analysis of randomized controlled trials. Diabetes Metabol Syndr Clin Res Rev 2017; 11(2): 157–62. doi: 10.1016/j.dsx.2016.06.025
- Liberati A, Altman DG, Tetzlaff J, Mulrow C, Gøtzsche PC, Ioannidis JP, et al. The PRISMA statement for reporting systematic reviews and meta-analyses of studies that evaluate health care interventions: explanation and elaboration. Ann Intern Med 2009; 151(4): W-65–W-94. doi: 10.7326/0003-4819-151-4-200908180-00136
- 24. Yang M, Bai W, Jiang B, Wang Z, Wang X, Sun Y, et al. Effects of a ketogenic diet in women with PCOS with different uric acid concentrations: a prospective cohort study. Reprod BioMed Online 2022; 45(2): 391–400. doi: 10.1016/j.rbmo.2022.03.023
- 25. Li J, Bai WP, Jiang B, Bai LR, Gu B, Yan SX, et al. Ketogenic diet in women with polycystic ovary syndrome and liver dysfunction who are obese: a randomized, open-label, parallel-group, controlled pilot trial. J Obstetr Gynaecol Res 2021; 47(3): 1145–52. doi: 10.1111/jog.14650
- 26. Avolio E, Ferraro S, Mocciaro R, De Pergola G, Buzzanca F, Venturella R, et al. The effect of ketogenic and low carb diet (Cyclic Alternation) on the ovarian morphology and insulin metabolism in polycystic ovary syndrome patients. Clin Obstet Gynecol 2020; 6: 1–6. doi: 10.15761/COGRM.1000307
- 27. Cincione RI, Losavio F, Ciolli F, Valenzano A, Cibelli G, Messina G, et al. Effects of mixed of a ketogenic diet in overweight and obese women with polycystic ovary syndrome. Int J Environ Res Public Health 2021; 18(23): 12490. doi: 10.3390/ ijerph182312490
- Cincione I, Graziadio C, Marino F, Vetrani C, Losavio F, Savastano S, et al. Short-time effects of ketogenic diet or modestly hypocaloric Mediterranean diet on overweight and obese women with polycystic ovary syndrome. J Endocrinol Investig 2023; 46(4): 769–77. doi: 10.1007/s40618-022-01943-y
- Mavropoulos JC, Yancy WS, Hepburn J, Westman EC. The effects of a low-carbohydrate, ketogenic diet on the polycystic ovary syndrome: a pilot study. Nutr Metabol 2005; 2(1): 1–5. doi: 10.1186/1743-7075-2-35
- 30. Liran B, Jian L, Litian N, Aili W, Fengyun W, Wenpei B, et al. Preliminary report of a single-center clinical study for ketogenic diet intervention of polycystic ovary syndrome. 实用临床医药杂 志. 2019; 23(7): 2–5.
- Jian L, Wenpei B, Bo J, Feiran L, Yanrong C. Effects of ketogenic diet on vitamin D and glycolipid metabolism in overweight or obese patients with polycystic ovary syndrome. 实用临床医药 杂志. 2022; 26(4): 14–17.
- 32. Pandurevic S, Mancini I, Mitselman D, Magagnoli M, Teglia R, Fazzeri R, et al. Efficacy of very low-calorie ketogenic diet with the Pronokal® method in obese women with polycystic ovary syndrome: a 16-week randomized controlled trial. Endocrine Connect 2023; 12(7): e220536. doi: 10.1530/EC-22-0536
- 33. Magagnini MC, Condorelli RA, Cimino L, Cannarella R, Aversa A, Calogero AE, et al. Does the ketogenic diet improve the quality of ovarian function in obese women? Nutrients 2022; 14(19): 4147. doi: 10.3390/nu14194147
- Basciani S, Camajani E, Contini S, Persichetti A, Risi R, Bertoldi L, et al. Very-low-calorie ketogenic diets with whey, vegetable,

or animal protein in patients with obesity: a randomized pilot study. J Clin Endocrinol Metabol 2020; 105(9): 2939–49. doi: 10.1210/clinem/dgaa336

- 35. Merra G, Gratteri S, De Lorenzo A, Barrucco S, Perrone M, Avolio E, et al. Effects of very-low-calorie diet on body composition, metabolic state, and genes expression: a randomized double-blind placebo-controlled trial. Eur Rev Med Pharmacol Sci 2017; 21(2): 329–45.
- 36. Saslow LR, Daubenmier JJ, Moskowitz JT, Kim S, Murphy EJ, Phinney SD, et al. Twelve-month outcomes of a randomized trial of a moderate-carbohydrate versus very low-carbohydrate diet in overweight adults with type 2 diabetes mellitus or prediabetes. Nutr Diabetes 2017; 7(12): 304. doi: 10.1038/ s41387-017-0006-9
- 37. Cunha GM, Guzman G, Correa De Mello LL, Trein B, Spina L, Bussade I, et al. Efficacy of a 2-month very low-calorie ketogenic diet (VLCKD) compared to a standard low-calorie diet in reducing visceral and liver fat accumulation in patients with obesity. Front Endocrinol 2020; 11: 607. doi: 10.3389/ fendo.2020.00607
- Zhao H, Jin H, Xian J, Zhang Z, Shi J, Bai X. Effect of ketogenic diets on body composition and metabolic parameters of cancer patients: a systematic review and meta-analysis. Nutrients 2022; 14(19): 4192. doi: 10.3390/nu14194192
- 39. Vargas S, Romance R, Petro JL, Bonilla DA, Galancho I, Espinar S, et al. Efficacy of ketogenic diet on body composition during resistance training in trained men: a randomized controlled trial. J Int Soc Sports Nutr 2018; 15(1): 31. doi: 10.1186/ s12970-018-0236-9
- Gomez-Arbelaez D, Bellido D, Castro AI, Ordoñez-Mayan L, Carreira J, Galban C, et al. Body composition changes after very-low-calorie ketogenic diet in obesity evaluated by 3 standardized methods. J Clin Endocrinol Metabol 2017; 102(2): 488–98. doi: 10.1210/jc.2016-2385
- 41. Luo W, Zhang J, Xu D, Zhou Y, Qu Z, Yang Q, et al. Low carbohydrate ketogenic diets reduce cardiovascular risk factor levels in obese or overweight patients with T2DM: a meta-analysis of randomized controlled trials. Front Nutr 2022; 9: 1092031. doi: 10.3389/fnut.2022.1092031
- Ibáñez L, de Zegher F. Adolescent PCOS: a postpubertal central obesity syndrome. Trends Mol Med 2023; 29(5): 354–363. doi: 10.1016/j.molmed.2023.02.006
- 43. Orio F, Muscogiuri G, Nese C, Palomba S, Savastano S, Tafuri D, et al. Obesity, type 2 diabetes mellitus and cardiovascular disease risk: an uptodate in the management of polycystic ovary syndrome. Eur J Obstetr Gynecol Reprod Biol 2016; 207: 214–9. doi: 10.1016/j.ejogrb.2016.08.026
- 44. Teede HJ, Misso ML, Deeks AA, Moran LJ, Stuckey BG, Wong JL, et al. Assessment and management of polycystic ovary syndrome: summary of an evidence-based guideline. Med J Aust 2011; 195(6): S65. doi: 10.5694/mja11.10915
- 45. Park Y-M, Zhang J, Steck SE, Fung TT, Hazlett LJ, Han K, et al. Obesity mediates the association between Mediterranean diet consumption and insulin resistance and inflammation in US adults. J Nutr 2017; 147(4): 563–571. doi: 10.3945/jn.116.243543

- 46. Zhang X, Zheng Y, Guo Y, Lai Z. The effect of low carbohydrate diet on polycystic ovary syndrome: a meta-analysis of randomized controlled trials. Int J Endocrinol 2019; 2019: 4386401. doi: 10.1155/2019/4386401
- 47. Teede H, Misso M, Costello M, Dokras A, Laven J, Moran L, et al. International PCOS Network. Recommendations from the international evidence-based guideline others Well-being and personality in PCOS for the assessment and management of polycystic ovary syndrome. Fertil Steril 2018; 110: 364–79. doi: 10.1016/j.fertnstert.2018.05.004
- Lin AW, Dollahite JS, Sobal J, Lujan ME. Health-related knowledge, beliefs and self-efficacy in women with polycystic ovary syndrome. Hum Reprod 2018; 33(1): 91–100. doi: 10.1093/ humrep/dex351
- Kozica SL, Gibson-Helm ME, Teede HJ, Moran LJ. Assessing self-efficacy and self-help methods in women with and without polycystic ovary syndrome. Behav Med 2013; 39(3): 90–6. doi: 10.1080/08964289.2012.720312
- Ye W, Xie T, Song Y, Zhou L. The role of androgen and its related signals in PCOS. J Cell Mol Med 2021; 25(4): 1825–37. doi: 10.1111/jcmm.16205
- Karpe F, Pinnick KE. Biology of upper-body and lower-body adipose tissue – link to whole-body phenotypes. Nat Rev Endocrinol 2015; 11(2): 90–100. doi: 10.1038/nrendo.2014.185
- Kim JY, Tfayli H, Michaliszyn SF, Arslanian S. Impaired lipolysis, diminished fat oxidation, and metabolic inflexibility in obese girls with polycystic ovary syndrome. J Clin Endocrinol Metabol 2018; 103(2): 546–54. doi: 10.1210/jc.2017-01958
- 53. Di Sarra D, Tosi F, Bonin C, Fiers T, Kaufman J-M, Signori C, et al. Metabolic inflexibility is a feature of women with polycystic ovary syndrome and is associated with both insulin resistance and hyperandrogenism. J Clin Endocrinol Metabol 2013; 98(6): 2581–8. doi: 10.1210/jc.2013-1161
- 54. Colica C, Merra G, Gasbarrini A, De Lorenzo A, Cioccoloni G, Gualtieri P, et al. Efficacy and safety of very-low-calorie ketogenic diet: a double blind randomized crossover study. Eur Rev Med Pharmacol Sci 2017; 21(9): 2274–89.
- 55. Marchetti M, Gualtieri P, Romano L, Merra G. What is the importance of saving lean mass in the treatment of obesity and related diseases? Eur Rev Med Pharmacol Sci 2019; 23(2): 431–2.
- 56. Colica C, Di Renzo L, Trombetta D, Smeriglio A, Bernardini S, Cioccoloni G, et al. Antioxidant effects of a hydroxyty-rosol-based pharmaceutical formulation on body composition, metabolic state, and gene expression: a randomized double-blinded, placebo-controlled crossover trial. Oxidat Med Cell Longevity 2017; 2017: 2473495. doi: 10.1155/2017/2473495

*Nan-nan Xing

Department of Obstetrics and Gynecology Shengjing Hospital of China Medical University Shenyang 110022, China Email: xingnn0114@sina.com